Author’s response to reviews

Title: Resting heart rate and impaired glucose regulation in Chinese middle-aged and elderly people: a cross-sectional analysis

Authors:

Zhen Yang (yangzhen1020@hotmail.com)
Weiwei Zhang (amanda_zww@126.com)
Xiaoyong Li (lixiaoyong@126.com)
Lingfei Zhu (zhuzhu@163.com)
Yixin Niu (niuniu@126.com)
Ning Lin (linning@163.com)
Shuai Lu (lushuai@163.com)
Hongmei Zhang (zanghongmei@126.com)
Xuanchun Wang (wangxunchun@126.com)
Jie Wen (wenjie@163.com)
Guang Ning (ningguang@126.com)
Li Qin (qinli@126.com)
Qing Su (suqingxinhua@163.com)

Version: 3 Date: 21 Aug 2017

Author’s response to reviews:

Dear Editor:

Our manuscript has been extensively revised in accordance with the enclosed reviewer’s comments. All changes in manuscripts emphasized by red characters.
- At the end of the discussion line 23 the sentence seems to suggest causality as critizised by the reviewers and this has not been addressed

Answers:

We thanks for editor’s reminder. This expression may be not rigorous. According to reviewer’s instruction, we have deleted this sentence in the manuscript.

- The manuscript still contains many mistakes (background, results as well as discussion) and would need thorough proofreading preferentially by a native speaker

Answers:

We thanks for editor’s reminder. According to editor’s suggestion, we have gotten language service from Nature Research Editing Service (http://bit.ly/NRES_BS) to improve English and flow.